Author's response to reviews

Title: Notch signaling in pediatric soft tissue sarcomas

Authors:

Rossella Rota (rossella.rota@opbg.net)
Roberta Ciarapica (roberta.ciarapica@yahoo.com)
Lucio Miele (lmiele@umc.edu)
Franco Locatelli (franco.locatelli@opbg.net)

Version: 3 Date: 10 October 2012

Author's response to reviews: see over
Roma, 10 of October, 2012

Dr. Sabina Alam  
Editor-in-chief, BMC Medicine  
BioMed Central Ltd,  
Floor 6, 236 Gray's Inn Road,  
London, WC1X 8HB,  
United Kingdom  
E-mail: bmcmedicineeditorial@biomedcentral.com

Dear Dr. Alam,

On behalf of the other authors I hereby resubmit a revised version of our Review entitled “Notch signaling in pediatric soft tissue sarcomas”.  
Please find enclosed a copy of the revised manuscript.  
Here we report a point by point description of our revisions.

Major revisions:

Please ensure that each author affiliation contains, in the following order, department, institution, street name and number, town, post code and country.  
We have checked and add what requested in the first page.

A figure 3 is cited on page 6, please correct this.  
We have eliminated the Figure 3 citation on page 6.

Please note that each reference can contain up to 30 authors' names for each citation. The term "et al." should not be used if there are fewer than 30 authors.  
We have edited the bibliography as requested.
Typography: we have checked our manuscript for typographical errors and made any necessary correction.

The number of words in the main text amounts to about 6100.

The contributing Authors listed below have agreed to the submission and are responsible for the contents of this manuscript.

Roberta Ciarapica, Laboratory of Angiogenesis, Dept. of Oncohematology, Ospedale Pediatrico Bambino Gesù, IRCCS, Piazza S. Onofrio 4, 00165 Roma, Italy.
E-mail: roberta.ciarapica@yahoo.com

Lucio Miele, Cancer Institute, University of Mississippi Medical Center, 2500 N. State Street, Guyton 2 Building, Suite G751-05, Jackson, MS 39216, USA.
E-mail: lmiele@umc.edu

Franco Locatelli, Dept. of Oncohematology, Clinical Oncology, Ospedale Pediatrico Bambino Gesù, IRCCS, Piazza S. Onofrio 4, 00165 Roma, Italy.
E-mail: franco.locatelli@opbg.net

We thank you for your attention and we are looking forward to hearing from you

Rossella Rota PhD
Laboratory of Angiogenesis
Dept. of Oncohematology
Ospedale Pediatrico Bambino Gesù
Piazza S. Onofrio 4
00165 Rome
Italy
rossella.rota@opbg.net
rossella_rota@yahoo.com
Ph:+39-06-6859.2648 office
Ph:+39-06-6859.2651/52 labs
Fax:+39-06-6859.2904/2101
mobile:+39-335-6175500